\documentclass[12pt,a4paper,draft]{amsart}
\usepackage{amsmath,amsthm}
\usepackage[english]{babel}
\usepackage{amsfonts}
\usepackage{latexsym}

\newcommand*{\cd}{(\cdot)}
\newcommand*{\lt}{L_2(\mathbb B^d)}
\newcommand*{\lp}{L_2([0,\pi])}
\newcommand*{\zs}{Z_{skj}}
\newcommand*{\ys}{Y_{skj}}
\newcommand*{\sk}{\sum_{s=1}^\infty\sum_{k=0}^\infty}
\newcommand*{\ja}{\sum_{j=1}^{a_k}}
\newcommand*{\wl}{\widehat\lambda}
\newcommand*{\wf}{\widehat f}
\newcommand*{\wu}{\widehat u}
\newcommand*{\wc}{\widehat c}
\newcommand*{\wb}{\widehat b}
\newcommand*{\as}{a_{sk}}

\newtheorem{theorem}{Theorem}


\begin{document}
\begin{thebibliography}{99}

\bibitem{TW}J.~F. Traub and H.~Wo\'zniakowski, {\it A General Theory of
Optimal Algorithms}, Academic Press, New York, 1980.

\bibitem{O}K.~Yu.~Osipenko, ``On recovery of the Dirichlet problem solution by inaccurate input data", {\it Vladikavkaz Mat. Zh.}, {\bf6}, 4 (2004), 55--62.

\bibitem{MOT}G.~G.~Magaril-Il'yaev, K.~Yu.~Osipenko, and V.~M.~Tikhomirov, ``On optimal recovery of heat equation solutions". In: Approximation
Theory: A volume dedicated to B. Bojanov (D.~K.~Dimitrov,
G.~Nikolov, and R.~Uluchev, Eds.), 163--175, Sofia: Marin Drinov
Academic Publishing House, 2004.

\bibitem{W}E.~V.~Wedenskaya, ``On optimal recovery of heat equation solution by inaccurate temperature given at several times" ,{\it Vladikavkaz Mat. Zh.}, {\bf8}, 1 (2006), 16--21.

\bibitem{B}E.~A.~Balova, ``On optimal recovery of the Dirichlet problem solution in an annulus, {\it Vladikavkaz Mat. Zh.}, {\bf8}, 2 (2006), 15--23.

\bibitem{VO}K.~Yu.~Osipenko and N.~D.~Vysk, ``Optimal recovery of wave equation solution by inaccurate input data", {\it Mat. Zametki} (2007) (to appear).

\bibitem{V}N.~D.~Vysk, ``On a wave equation solution with inaccurate Fourier
coefficients of function
defined the initial form of string", {\it Vladikavkaz Mat. Zh.}, {\bf8}, 4 (2006), 12--17.

\bibitem{MO}G.~G. Magaril-Il'yaev and K.~Yu.~Osipenko, ``Optimal recovery of functions and their derivatives from Fourier coefficients prescribed with an error", {\it Mat. Sb.} {\bf193} (2002), 79--100; English transl. in {\it Sb. Math}. {\bf193} (2002).

\bibitem{MO1}G.~G. Magaril-Il'yaev and K.~Yu.~Osipenko, ``Optimal recovery of functions and their derivatives from inaccurate information about a
spectrum and inequalities for derivatives", {\it Funkc. analiz i ego
prilozh.} {\bf37} (2003), 51--64; English transl. in {\it Funct. Anal and
Its Appl.}, {\bf37} (2003).

\bibitem{Os}K.~Yu.~Osipenko, ``The Hardy--Littlewood--P\'olya inequality for analytic functions from Hardy--Sobolev spaces",  {\it Mat. Sb.} {\bf197} (2006), 15--34; English transl. in {\it Sbornic: Mathematics} {\bf197} (2006), 315--334.

\bibitem{MM}A.~A.~Melkman and C.~A.~Micchelli, ``Optimal estimation of linear operators in Hilbert spaces from inaccurate data", {\it SIAM J. Numer. Anal.}, {\bf16} (1979), 87--105.

\bibitem{Pl}L.~Plaskota, {\it Noisy Information and Computational complexity}, Cambridge University Press, 1996.

\bibitem{We}A.~G.~Werschulz, {\it The Computational Complexity of Differential and Integral Equations: An Information-Based Approach}, Oxford University Press, New York, 1991.


\bibitem{St}E.~M.~Stein and G.~Weiss, {\it Introduction to Fourier analysis on Euclidean spaces}, Princeton University Press, Princeton, NJ 1971.

\bibitem{Ab}M.~Abramovitz and I.~A.~Stegun, {\it Handbook of Mathematical Functions}, Dover, New York, 1972.

\end{thebibliography}


\title[Optimal Recovery of Solutions of the Heat Equation]{Optimal Recovery of Solutions of the Generalized Heat Equation in the Unit Ball from Inaccurate Data}
\author{K.~Yu. Osipenko}\address{``MATI" --- Russian State Technological University}\email{konst@yahoo.com}
\author{E.~V.~Wedenskaya}\address{``MATI" --- Russian State Technological University}\email{wed@rstu.ru}

\thanks{The research was carried out with the financial support of the
Russian Foundation for Basic Research (grant nos.\ 05-01-00275, 06-01-81004, 05-01-00261, and 06-01-00530) and the President Grant for State Support of Leading Scientific Schools in Russian Federation (grant                        no. NSH-5813.2006.1)}

\begin{abstract}
We consider the problem of optimal recovery of solutions of the generalized heat equation in the unit ball. Information is given at two time instances, but inaccurate. The solution is to be constructed at some intermediate time. We provide the optimal error and present an algorithm which achieves this error level.
\end{abstract}

\dedicatory{We devote this paper to Henryk Wo\'zniakowski who had intensively studied optimal recovery problems}

\
\maketitle

The application of optimal recovery theory to problems of partial differential equations was started by J.~F.~Traub and H.~Wo\'zniakowski in \cite{TW}. In particular, this monograph considered optimal recovery of solutions of the heat equation from finitely many Fourier coefficients of the initial function. Several recovery problems for partial differential equation from noisy information were recently studied in \cite{O}--\cite{V}. The results considered in these papers were based on a general method for optimal recovery of linear operators developed in \cite{MO} and \cite{MO1} (see also \cite{Os}).
This method extended previous research from \cite{MM}. Various problems of optimal recovery from noisy information may be found in \cite{Pl} (see also \cite{We} where the complexity of differential and integral equations is discussed).

Here we consider the optimal recovery problem for solutions of the generalized heat equation in the unit $d$-ball at the time $\tau$ from inaccurate solutions at the times $t_1$ and $t_2$.

Set
\begin{gather*}
\mathbb B^d=\biggl\{\,x=(x_1,\dots,x_d):|x|^2=\sum_{j=1}^d x_j^2<1\,\biggr\},\\
\mathbb S^{d-1}=\{\,x\in\mathbb R^d:|x|=1\,\}.
\end{gather*}
Consider the problem of finding the solution of the generalized heat equation in $\lt$:
\begin{equation}\label{1}
\begin{aligned}
&u_t+(-\Delta)^{\alpha/2}u=0,\quad\alpha>0,\\
&u|_{t=0}=f(x),\\
&u|_{x\in\mathbb S^{d-1}}=0.
\end{aligned}
\end{equation}

Let $0\le t_1<t_2$. Suppose we know approximate solutions $y_1$ and $y_2$ of \eqref{1} at times $t_1$ and $t_2$, given with errors $\delta_1$ and $\delta_2$ in the $\lt$ norm. We want to recover in the best way the solution of \eqref{1} at the time $\tau$ , $t_1<\tau<t_2$. We assume that
$y_1,y_2\in\lt$ satisfy
$$\|u(\cdot,t_j)-y_j\cd\|_{\lt}\le\delta_j,\quad j=1,2.$$
Any map $\xi\colon\lt\times\lt\to\lt$ is admitted as a recovery method. The quantity
\begin{multline*}
e_\tau(\alpha,\lt,\delta_1,\delta_2,\xi)\\
=\sup_{\substack{f,y_1,y_2\in\lt\\
\|u(\cdot,t_j)-y_j\cd\|_{\lt}\le\delta_j,\ j=1,2}}
\|u(\cdot,\tau)-\xi(y_1,y_2)\cd\|_{\lt},
\end{multline*}
where $u$ is the solution of \eqref{1}, is called the error of the method $\xi$. The quantity
$$E_\tau(\alpha,\lt,\delta_1,\delta_2)=
\inf_{\xi\colon\lt\times\lt\to\lt}e_\tau(\alpha,\lt,\delta_1,\delta_2,\xi)$$
is called the error of optimal recovery and a method delivering the lower bound is called an optimal recovery method.

Note that the initial functions $f$ belong to the whole space $\lt$. In other words, the a priori information about initial functions is not a compact set. Therefore we use the information with infinite cardinality (\cite{TW} dealt with algorithms using information having finite cardinality). For example, it can be shown that knowing (even precisely) any finite number of Fourier coefficients of $u(\cdot,t_j)$, $j=1,2$, does not lead to the finite error of optimal recovery.

The analysis of the problem is different for $d=1$ and $d>1$, because of different type of orthogonal eigensystems.

We begin with the case $d>1$. Let $H_k$ denote the set of spherical harmonics of order $k$. It is known
(see \cite{St}) that $\dim H_0=a_0=1$,
$$\dim H_k=a_k=(d+2k-2)\frac{(d+k-3)!}{(d-2)!k!},\quad k=1,2,\ldots,$$
and
$$L_2(\mathbb S^{d-1})=\sum_{k=0}^\infty H_k.$$

Let $\{Y_j^{(k)}\}_{j=1}^{a_k}$ denote an orthonormal basis in $H_k$. Let $J_p$ be the Bessel function of the first kind of order $p$ and $\mu_s^{(p)}$, $s=1,2,\ldots,$ be the zeros of $J_p$.

The functions
$$\zs(x)=\frac{J_p(\mu_s^{(p)}r)}{r^{d/2-1}}Y_j^{(k)}(x'),$$
where $r=|x|$, $x'=x/r$, and $p=k+(d-2)/2$, form an orthogonal basis in $\lt$. Moreover,
$$\Delta \zs=-(\mu_s^{(p)})^2 \zs.$$
We will use the orthonormal basis in $\lt$
$$\ys=\frac{\zs}{\|\zs\|_{\lt}}.$$

We recall that the operator $(-\Delta)^{\alpha/2}$ is defined as follows
$$(-\Delta)^{\alpha/2}f=\sk(\mu_s^{(p)})^\alpha\ja c_{skj}\ys,$$
where
\begin{equation}\label{Four}
f=\sk\ja c_{skj}\ys.
\end{equation}

The solution of \eqref{1} can be easily found by the Fourier method of separation of variables. It has the form
$$u(x,t)=\sk e^{-(\mu_s^{(p)})^\alpha t}\ja c_{skj}Y_{skj}(x),$$
where $c_{skj}$ are the Fourier coefficients of the initial function.

Set
$$\as=e^{-2{(\mu_s^{(p)})}^\alpha}$$
(we recall that $p=k+(d-2)/2$ and $\alpha$ is from \eqref{1}).
It is known (see~\cite{Ab}) that for all $s\in\mathbb N$
$$\mu_s^{(p)}<\mu_s^{(p+1)}<\mu_{s+1}^{(p)}$$
and $\mu_s^{(p)}\to\infty$ as $s\to\infty$. So the set of zeros of the Bessel functions $\mu_s^{(p)}$, $s=1,2,\ldots$, $p=k+(d-2)/2$, $k=0,1,\ldots,$ can be arranged in ascending order
$$\mu_{s_1}^{({p_1})}<\mu_{s_2}^{({p_2})}<\ldots<\mu_{s_n}^{(p_n)}<\ldots\,.$$
Consequently,
$$a_{s_1k_1}>a_{s_2k_2}>\ldots>a_{s_nk_n}>\ldots\,.$$

For the case $d=1$ the functions
$$Y_s(x)=\sin\frac{\pi s}2(x+1),\quad s=1,2,\ldots,$$
form an orthonormal basis in $L_2(\mathbb B^1)=L_2([-1,1])$ and
$$\Delta Y_s=-\left(\frac{\pi s}2\right)^2Y_s.$$
We define the operator $(-\Delta)^{\alpha/2}$ as follows
$$(-\Delta)^{\alpha/2}f=\sum_{s=1}^\infty\left(\frac{\pi s}2\right)^\alpha c_sY_s,$$
where $c_s$ are the Fourier coefficients of $f$. It is easily verified that for $d=1$ the solution of \eqref{1} is given by
$$u(x,t)=\sum_{s=1}^\infty e^{-\left(\frac{\pi s}2\right)^\alpha t}c_sY_s(x),$$
where $c_s$ are the Fourier coefficients of the initial function.

For an arbitrary decreasing sequence $\beta_1>\beta_2>\ldots>0$ we introduce the following notation
\begin{align*}
\Delta_m& =\left[\beta_{m+1}^{t_2-t_1},\beta_m^{t_2-t_1}\right],\quad\Delta_0=\left[
\beta_1^{t_2-t_1},+\infty \right),\\
\wl_1&=\begin{cases}\displaystyle\frac{\beta_{m+1}^{\tau-t_2}-{\beta_m}^{\tau-t_2
}}{\beta_{m+1}^{t_1-t_2}-\beta_m^{t_1-t_2}},&\dfrac{\delta_2^2}{\delta_1^2}
\in\Delta_m,\ m\ge1,\\[12pt]
\beta_1^{\tau-t_1},&\dfrac{\delta_2^2}{\delta_1^2}\in\Delta_0,
\end{cases}\\
\wl_2&=\begin{cases}\displaystyle\frac{\beta_m^{\tau-t_1}-\beta_{m+1}^{\tau-t_1}}
{\beta_m^{t_2-t_1}-\beta_{m+1}^{t_2-t_1}}&\dfrac{\delta_2^2}{\delta_1^2}
\in\Delta_m,\ m\ge1,\\[12pt]
0,&\dfrac{\delta_2^2}{\delta_1^2}\in\Delta_0.
\end{cases}
\end{align*}

\begin{theorem}\label{T1}
Set
$$\beta_m=\begin{cases}a_{s_m,k_m},&d>1,\\[8pt]
e^{-2\left(\frac{\pi m}2\right)^\alpha},&d=1.\end{cases}$$
Then for all $\delta_1,\delta_2>0$ the following equality
$$E_\tau(\alpha,\lt,\delta_1,\delta_2)=\sqrt{\wl_1\delta_1^2+\wl_2\delta_2^2}$$
holds. Moreover, the method
\begin{equation}\label{met}
\widehat{\xi}(y_1,y_2)(x)=\begin{cases}\displaystyle\sk\as^{\tau/2}\ja\frac{\wl_1\as^{t_1/2}y_{1skj
}+\wl_2\as^{t_2/2}y_{2skj}}{\wl_1\as^{t_1}+\wl_2\as^{t_2}}Y_{skj}(x),\\
\hspace{230pt}d>1,\\
\displaystyle\sum_{s=1}^\infty e^{-\left(\frac{\pi s}2\right)^\alpha\tau}\frac{\wl_1e^{-\left(\frac{\pi s}2\right)^\alpha t_1}y_{1s}+\wl_2e^{-\left(\frac{\pi s}2\right)^\alpha t_2}y_{2s}}
{\wl_1e^{-2\left(\frac{\pi s}2\right)^\alpha t_1}+\wl_2e^{-2\left(\frac{\pi s}2\right)^\alpha t_2}}Y_s(x),\\
\hspace{230pt}d=1,\end{cases}
\end{equation}
where $y_{1skj},y_{2skj}$ and $y_{1s},y_{2s}$ are the Fourier coefficients of $y_1\cd$ and $y_2\cd$, is optimal.
\end{theorem}

To prove Theorem~\ref{T1} we use a general scheme of construction of optimal recovery methods for linear operators developed in
\cite{MO} and \cite{MO1} (see also \cite{Os}).

Consider the following extremal problem
\begin{multline}\label{ex}
\|u(\cdot,\tau)\|_{\lt}^2\to\max,\quad\|u(\cdot,t_j)\|_{\lt}^2\le\delta_j^2,\ j=1,2,\\
f\in\lt,
\end{multline}
where $u$ is the solution of problem \eqref{1}. Set
$$L(f,\lambda_1,\lambda_2)=-\|u(\cdot,\tau)\|_{\lt}^2+\lambda_1\|u(\cdot,t
_1)\|_{\lt}^2+\lambda_2\|u(\cdot,t_2)\|_{\lt}^2.$$


From \cite{MO1} (see also \cite{Os}) follows

\begin{theorem}\label{T2}
Suppose that there exist $\wl_1\ge0$, $\wl_2\ge0$ and an admissible  function $\wf$ in \eqref{ex} such that
\begin{align*}
(a)&\quad\min_{f\in\lt}L(f,\wl_1,\wl_2)=L(\wf,\wl_1,\wl_2),\\
(b)&\quad\wl_1(\|\wu(\cdot,t_1)\|_{\lt}^2-\delta_1^2)+\wl_2(\|\wu(\cdot,t_2)
\|_{\lt}^2-\delta_2^2)=0,
\end{align*}
where $\wu$ is the solution of \eqref{1} with the initial function $\wf$. If for all $y_1,y_2\in\lt$
there exists a solution $f_0$ of the problem
\begin{multline*}
\wl_1\|u(\cdot,t_1)-y_1\cd\|_{\lt}^2+\wl_2\|u(\cdot,t_2)-y_2\cd\|_{\lt}^2\to\min,\\ f\in\lt,
\end{multline*}
where $u$ is the solution of \eqref{1},
then the method
$$\widehat{\xi}(y_1,y_2)(x)=u_0(x,\tau),$$
where $u_0$ is the solution of \eqref{1} with the initial function $f_0$, is optimal and for the error of optimal recovery the following equality
$$E(\alpha,\lt,\delta_1,\delta_2)=\sqrt{\wl_1\delta_1^2+\wl_2\delta_2^2}$$
holds.
\end{theorem}

\begin{proof}[Proof of Theorem~$\ref{T1}$]
Consider the case $d>1$. We have
$$L(f,\wl_1,\wl_2)=\sk(-\as^\tau+\wl_1\as^{t_1}+\wl_2\as^{t_2})\ja c_{skj}^2,$$
where $c_{skj}$ are the Fourier coefficients of $f$. Putting
$$b_{sk}=\ja c_{skj}^2,$$
we rewrite $L(f,\wl_1,\wl_2)$ in the form
$$L(f,\wl_1,\wl_2)=\sk\as^\tau(-1+\wl_1\as^{t_1-\tau}
+\wl_2\as^{t_2-\tau})b_{sk}.$$
Assume that $\delta_2^2/\delta_1^2\in\Delta_m$, $m\ge1$. It is easily seen that in this case for $\wl_1$ and $\wl_2$, equalities
\begin{equation}\label{eql}
\begin{aligned}
\wl_1\beta_m^{t_1}+\wl_2\beta_m^{t_2}&=\beta_m^\tau,\\
\wl_1\beta_{m+1}^{t_1}+\wl_2\beta_{m+1}^{t_2}&=\beta_{m+1}^\tau
\end{aligned}
\end{equation}
hold. Consider the function $g(z)=-1+\wl_1 e^{-2z(t_1-\tau)}+\wl_2 e^{-2z(t_2-\tau)}$. It is easy to verify that $g$ is a convex function. It follows from \eqref{eql} that $g$ has two zeros $z_m={({\mu_{s_m}}^{(p_m)})}^\alpha$ and $z_{m+1}={({\mu_{s_{m+1}}}^{(p_{m+1})})}^\alpha$. In view of the convexity of $g$ for all $z\le z_m$ and all $z\ge z_{m+1}$ the inequality $g(z)\ge0$ holds. Thus for all $f\in\lt$ we have
$$L(f,\wl_1,\wl_2)\ge0.$$
Define $\wb_{s_m,k_m}$ and $\wb_{s_{m+1},k_{m+1}}$ from the conditions
$$\wb_{s_m,k_m}\beta_m^{t_j}+\wb_{s_{m+1},k_{m+1}}\beta_{m+1}^{t_j}=\delta_j^2,\quad j=1,2.$$
It is easy to verify that
\begin{align*}
\wb_{s_m,k_m}&=\frac{\delta_1^2}{\beta_m^{t_1}}\cdot
\frac{\delta_2^2/\delta_1^2-\beta_{m+1}^{t_2-t_1}}{\beta_m^{t_2-t_1}
-\beta_{m+1}^{t_2-t_1}},\\
\wb_{s_{m+1},k_{m+1}}&=\frac{\delta_1^2}{\beta_{m+1}^{t_1}}\cdot
\frac{\beta_m^{t_2-t_1}-\delta_2^2/\delta_1^2}{\beta_m^{t_2-t_1}
-\beta_{m+1}^{t_2-t_1}}.
\end{align*}
For $j\ne m,m+1$ we set $b_{s_j,k_j}=0$. Then the function
$$\wf(x)=\sum_{j=m}^{m+1}\sqrt{b_{s_jk_j}}Y_{s_jk_j1}(x)$$
will be admissible and
$$L(\wf,\wl_1,\wl_2)=0.$$
Thus conditions $(a)$ and $(b)$ of Theorem~\ref{T2} hold.

Now we assume that $\delta_2^2/\delta_1^2\in\Delta_0$. It means that $\delta_2^2\ge\delta_1^2\beta_1^{t_2-t_1}$. Putting
$$\wf(x)=\delta_1\beta_1^{-t_1/2}Y_{s_1k_11}(x),$$
for the solution $\wu$ of \eqref{1} with the initial function $\wf$ we have
\begin{align*}
\|\wu(\cdot,t_1)\|_{\lt}^2&=\delta_1^2,\\
\|\wu(\cdot,t_2)\|_{\lt}^2&=\delta_1^2\beta_1^{t_2-t_1}\le\delta_2^2.
\end{align*}
Consequently, condition $(b)$ of Theorem~\ref{T2} holds. Condition $(a)$ of the same theorem holds since for all functions $f\in\lt$ $$L(f,\wl_1,\wl_2)\ge0,$$
and moreover
$$L(\wf,\wl_1,\wl_2)=0.$$

Now let us construct an optimal recovery method. According to Theorem~\ref{T2} we have to solve the problem
\begin{multline*}
\wl_1\sk\ja{(\as^{t_1/2}c_{skj}-y_{1skj})}^2\\
+\wl_2\sk\ja\left(\as^{t_2/2}c_{skj}-y_{2skj}\right)^2\to\min,\quad
f\in\lt,
\end{multline*}
where $c_{skj}$ are the Fourier coefficients of $f$ (see~\eqref{Four}).
It can be easily verified that the solution of this problem has the form
$$\widehat c_{skj}=\frac{\wl_1\as^{t_1/2}y_{1skj}+\wl_2\as^{t_2/2} y_{2skj}}{\wl_1\as^{t_1}+\wl_2\as^{t_2}}.$$
The optimality of method \eqref{met} now follows from Theorem~\ref{T2}.

The case $d=1$ may be considered in a similar way.
\end{proof}

We give the table (see~\cite{Ab}) of the first ten ordered numbers $\mu_s^{(p)}$ for even $d$ (that is, for $p\in\mathbb Z_+$) and for odd $d$ (when $p=k+1/2$, $k=0,1,\ldots$).

$$\begin{array}{|r|c|c|c||c|c|c|}
\hline
\multicolumn{1}{|c|}{j}&\rule{0pt}{20pt}s_j&p_j&\mu_{s_j}^{(p_j)}&s_j&p_j&\mu_{s_j}^{(p_j)}\\[10pt]
\hline
\rule{0pt}{20pt}1&1&0&2.4048&1&\dfrac12&3.1416\\[10pt]
\hline
\rule{0pt}{20pt}2&1&1&3.8317&1&\dfrac32&4.4934\\[10pt]
\hline
\rule{0pt}{20pt}3&1&2&5.1356&1&\dfrac52&5.7635\\[10pt]
\hline
\rule{0pt}{20pt}4&2&0&5.5200&2&\dfrac12&6.2832\\[10pt]
\hline
\rule{0pt}{20pt}5&1&3&6.3802&1&\dfrac72&6.9879\\[10pt]
\hline
\rule{0pt}{20pt}6&2&1&7.0156&2&\dfrac32&7.7253\\[10pt]
\hline
\rule{0pt}{20pt}7&1&4&7.5883&1&\dfrac92&8.1826\\[10pt]
\hline
\rule{0pt}{20pt}8&2&2&8.4172&2&\dfrac52&9.0950\\[10pt]
\hline
\rule{0pt}{20pt}9&3&0&8.6537&1&\dfrac{11}2&9.3558\\[10pt]
\hline
\rule{0pt}{20pt}10&1&5&8.7715&3&\dfrac12&9.4248\\[10pt]
\hline
\end{array}$$

The authors are grateful to referees for their remarks and suggestions which greatly help us to improve the paper.



\end{document}
